Author’s response to reviews

Title: POST - EFFECTS OF OBSTETRIC FISTULA IN UGANDA; A CASE STUDY OF FISTULA SURVIVORS IN KITOBU MISSION HOSPITAL (MASAKA), UGANDA

Authors:

John Bosco Bomboka (bombokajb@gmail.com)
Mary Gorrethy Nalubega-Mboowa (mgmboowa03@gmail.com)
Jennifer Nakilembe (jnakilembe@gmail.com)

Version: 3 Date: 14 May 2019

Author’s response to reviews:

I have responded to all the specific reviewer and editor comments (1 - 6)

1. We note that the current submission contains some textual overlap with other previously published works,

Please re-phrase these sections to minimise overlap. - The write up in these sections have been revised and re-phrased under the discussion section pages (11 - 15)

2. The sections that were requested to be added in the “Ethics approval and consent to participate” section of my declarations have been added... pages 16 - 17

3. Please clearly indicate the corresponding author on the title page.

- This has been done on the title page (1)

4. Please clarify whether the interview guide used in your study was developed for this study or has previously been published elsewhere.

- The interview guide had been included in the manuscript under the Additional files section page 21 and also uploaded under supplementary files

5. Please have the text edited by a professional language editing service or a native English speaking colleague.

- The manuscript was submitted to a professional language editor to sort issues with grammar, wording, spelling, and/or punctuation
6. The revised manuscript has been uploaded with supplementary files and all necessary files referenced in the manuscript text as required